# UNIVERSITY BIRMINGHAM University of Birmingham Research at Birmingham

# Impact of coronavirus disease 2019 on patients with primary adrenal insufficiency

Knowles, Gregory; Warmington, Emily; Shepherd, Lisa M; Hazlehurst, Jonathan M; de Bray, Anne; Gleeson, Helena; Arlt, Wiebke; Prete, Alessandro

DOI: 10.1530/EC-23-0122

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

## Citation for published version (Harvard):

Knowles, G, Warmington, E, Shepherd, LM, Hazlehurst, JM, de Bray, A, Gleeson, H, Arlt, W & Prete, A 2023, 'Impact of coronavirus disease 2019 on patients with primary adrenal insufficiency: a cross-sectional study', *Endocrine Connections*, vol. 12, no. 8, e230122. https://doi.org/10.1530/EC-23-0122

Link to publication on Research at Birmingham portal

#### **General rights**

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

#### Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



**12**:8

# Impact of coronavirus disease 2019 on patients with primary adrenal insufficiency: a cross-sectional study

Gregory Knowles<sup>1,\*</sup>, Emily Warmington<sup>2,\*</sup>, Lisa M Shepherd<sup>3,4</sup>, Jonathan M Hazlehurst<sup>3,5</sup>, Anne de Bray<sup>3,4</sup>, Helena Gleeson<sup>3</sup>, Wiebke Arlt<sup>3,4,6</sup> and Alessandro Prete<sup>03,4,7</sup>

<sup>1</sup>Walsall Manor Hospital, Walsall, UK

<sup>2</sup>College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

<sup>3</sup>Department of Endocrinology, University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

<sup>4</sup>Institute of Metabolism and Systems Research, University of Birmingham, Birmingham, UK

<sup>5</sup>Institute of Applied Health Research, University of Birmingham, Birmingham, UK

<sup>6</sup>Medical Research Council London Institute of Medical Sciences, London, UK

<sup>7</sup>NIHR Birmingham Biomedical Research Centre, University of Birmingham and University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

#### Correspondence should be addressed to A Prete: A.Prete@bham.ac.uk

\*(G Knowles and E Warmington contributed equally to this work)

# Abstract

*Objective:* Patients with primary adrenal insufficiency (PAI) are thought to be particularly vulnerable to coronavirus disease 2019 (COVID-19); however, little is known about its true impact on this group. We assessed morbidity and health promotion attitudes during the pandemic amongst a large cohort of patients with PAI.

*Design:* Cross-sectional, single-centre study.

*Methods:* In May 2020, COVID-19 advice on social distancing and sick-day rules was distributed to all patients with PAI registered with a large secondary/tertiary care centre. A semi-structured questionnaire was used to survey patients in early 2021.

*Results:* Of 207 contacted patients, 162 responded (82/111 with Addison's disease, AD; 80/96 with congenital adrenal hyperplasia, CAH). Patients with AD were older than those with CAH (median age 51 vs 39 years; P < 0.001) and had more comorbidities (Charlson comorbidity index  $\ge 2$  47.6% vs 10.0%; P < 0.001). By the time of the survey, 47 patients (29.0%) had been diagnosed with COVID-19, the second commonest cause of sick-day dosing during the study and the leading trigger of adrenal crises (4/18 cases). Patients with CAH had a higher risk of COVID-19 compared to AD (adjusted odds ratio 2.53 (95% CI 1.07–6.16), P = 0.036), were less inclined to have the COVID-19 vaccine (80.0% vs 96.3%; P = 0.001), and were less likely to have undergone hydrocortisone self-injection training (80.0% vs 91.5%; P = 0.044) or wear medical alert jewellery (36.3% vs 64.6%; P = 0.001).

*Conclusions:* COVID-19 was a principal trigger for adrenal crises and sick-day dosing in patients with PAI. Despite a higher risk of COVID-19, patients with CAH showed less engagement with self-protective attitudes.

*Significance statement:* We conducted a cross-sectional study on a large and well-characterised group of patients with PAI and demonstrated that COVID-19 was a leading cause of morbidity during the early phases of the pandemic. Patients with AD were older and had a greater burden of comorbidity than those with CAH, including non-adrenal autoimmune disorders. However, patients with CAH were more likely to develop COVID-19 and demonstrated reduced engagement with healthcare services and health promotion strategies.

#### **Key Words**

- COVID-19
- SARS CoV 2
- ► coronavirus
- Addison's disease
- congenital adrenal hyperplasia
- adrenal insufficiency
- surveys and questionnaires

Endocrine Connections (2023) **12**, **e230122** 





# Introduction

Primary adrenal insufficiency (PAI) is characterised by inadequate steroid hormone secretion caused by intrinsic adrenal cortex disease (1). The most common causes of PAI are Addison's disease (AD), usually resulting from autoimmune destruction of the adrenal cortex, and congenital adrenal hyperplasia (CAH), a group of autosomal recessive inborn disorders caused by enzyme deficiencies in adrenal steroidogenesis (2). PAI necessitates long-term glucocorticoid replacement, the dosage of which should be increased under stressors, such as infection, trauma or surgery, to prevent life-threatening adrenal crises, with increased stress doses recommended for patients with PAI and COVID-19 (1, 2, 3). Despite advancements in knowledge of the condition and novel treatment strategies, patients with PAI still face increased morbidity and mortality risk. Most notably, several studies have proposed that these patients may carry an increased risk of infections, which are a major trigger for adrenal crises (4, 5, 6, 7, 8, 9, 10, 11, 12).

Coronavirus disease 2019 (COVID-19) is a respiratory infection caused by the novel severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2). Presentation of COVID-19 ranges from asymptomatic infection to acute respiratory distress syndrome, multiorgan failure and death. One key negative prognostic factor for severe COVID-19 is compromised immunity and increased infection risk (13); as such, patients with PAI are assumed to be particularly vulnerable to COVID-19 (9, 14). However, due to the novelty of the disease, little is known about its true impact on this patient group. A case-control study including 60 patients with PAI from an area in Italy at the epicentre of a COVID-19 outbreak reported the prevalence of symptoms suggestive of COVID-19 as similar between those with and without adrenal insufficiency, with no patients requiring hospitalisation or experiencing adrenal crises (15). A survey of 40 patients with PAI assessing the impact of the pandemic on physical and psychological health identified only one case of COVID-19, also reported as mild (16). A more recent longitudinal study exploring self-reported COVID-19-related outcomes found that patients with adrenal insufficiency perceived difficulties accessing medical care and self-managing their condition during the pandemic, resulting in feelings of anxiety and reduced quality of life (17).

Since patients with PAI may face an increased susceptibility to infections (9, 14) and there is some suggestion that patients have struggled with self-management of their disease during the pandemic, we

aimed to assess the impact of COVID-19 on morbidity and health promotion attitudes of a large cohort of patients with PAI.

# **Materials and methods**

## Participants

Patients diagnosed with AD and CAH under the care of the Endocrinology Department at the Queen Elizabeth Hospital Birmingham and Heartlands Hospital (University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK) were identified from electronic medical records. In May 2020, when the first lockdown was starting to be lifted in England, a letter containing COVID-19 information was distributed to all patients. The letter highlighted the increased risk of patients with PAI when contracting COVID-19 and potentially developing serious complications and contained advice on strict social distancing and sick day rules (Appendix Document 1, see section on supplementary materials given at the end of this article). Patients were also signposted to the latest government advice and provided with contact details for the hospital endocrinology team.

#### Questionnaire and outcomes measured

A semi-structured telephone questionnaire was used to survey patients with PAI between January and April 2021 (Appendix Document 2). At this time, England had entered a third national lockdown and COVID-19 vaccinations were beginning to be rolled out, with patients prioritised for vaccination according to their age and medical history.

We gathered both demographic and clinical data; where available, clinical information was verified using electronic medical records. The Charlson Comorbidity Index was used as a measure of comorbidities to predict mortality (18). Patients were asked about their history of suspected (presence of typical symptoms (a high temperature, a new continuous cough or a loss or change to the sense of smell or taste) without polymerase chain reaction (PCR) swab confirmation) or confirmed (positive PCR swab) COVID-19 and clinical sequelae, illness within the preceding 12 months, knowledge of adrenal crisis prevention strategies as well as health promotion attitudes including ideas around the COVID-19 vaccination. PCRpositive COVID-19 was defined as severe if requiring hospital admission or triggering an adrenal crisis. An adrenal crisis was diagnosed if a patient presented with at least two of the following: hypotension or hypovolaemic







Figure 1

Flowchart of participant inclusion. A diagram outlining our patient recruitment process. 162 patients were included in the analysis. Patients have been categorised into their disease aetiology. Abbreviations: 21OH, 21-hydroxylase; 11BOH, 11-beta-hydroxylase; 3BHSD, 3-betahydroxysteroid dehydrogenase.

shock; nausea or vomiting; severe fatigue; fever; impaired consciousness (19).

Institutional review board approval of the questionnaire and its use was obtained by the University Hospitals Birmingham National Health Service (NHS) Foundation Trust, UK (CARMS-16750).

# **Statistical analysis**

Categorical data were expressed as counts and percentages, and the Fisher's exact test was applied to assess differences between groups. Quantitative variables were reported as median and interquartile range (IQR), and the Mann-Whitney *U* test was used for comparisons. Logistic regression was used to assess the odds of diagnosed or suspected COVID-19 in patients with CAH compared to those with AD. A multivariate model was used to generate an adjusted odds ratio using age, sex, body mass index (BMI), type of glucocorticoid used, daily hydrocortisoneequivalent dose, fludrocortisone treatment, Charlson Comorbidity Index score, smoking status, work and living arrangements and ethnicity as covariates. *P*-values <0.05 were considered statistically significant. The statistical



analyses were carried out using GraphPad Prism 9 (GraphPad Software).

# Results

# **Patient characteristics**

Of 207 patients with PAI approached, 45 (AD, N = 29; CAH, N = 16) either declined to participate in the telephone interview or could not be contacted (78.3% response rate). The electronic medical records of these 45 patients were reviewed and no deaths or COVID-19-related hospital admissions were identified.

Of the 162 patients included in the analysis, 82 had a diagnosis of AD and 80 of CAH (Fig. 1). The most common cause of CAH was 21-hydroxylase deficiency, accounting for 96.3% of cases. Three patients with AD had a diagnosis of autoimmune polyendocrine syndrome type 1.

Most patients were women (AD, 62.2%; CAH, 64.4%). At the time of the telephone interview, the median age was 55 years (IQR 39–67) and 39 years (IQR 29–47) in patients with AD and CAH, respectively (Table 1). Both groups were similar in terms of ethnicity and BMI.



 Table 1
 Baseline characteristics of included patients.

	Addison's disease (n = 82)	<b>Congenital adrenal hyperplasia</b> ( <i>n</i> = 80)	<i>P</i> -value
Women. <i>n</i> (%)	51 (62.2)	50 (64.4)	>0.999
Age (vears), median (IOR) <sup>a</sup>	55 (39–67)	39 (29–47)	< 0.001
Ethnicity. n (%)		( - )	
White	73 (89.0)	62 (77.5)	0.059
Asian or Asian British	6 (7.3)	14 (17.5)	0.058
Black, African, Caribbean, or Black British	2 (2.4)	4 (5.0)	0.440
Other ethnic group	1 (1.2)	_	>0.999
BMI (kg/m <sup>2</sup> ), median (IQR)	25.7 (23.2–30.0)	27.9 (22.3-31.8)	0.449
Years since diagnosis, median (IQR)	16 (8–28)	_	-
Comorbidities			
Autoimmune diseases, <i>n</i> (%)ª	35 (42.7)	8 (10.0)	< 0.001
Hypothyroidism, <i>n</i>	26	1	
Type 1 diabetes mellitus, <i>n</i>	12	1	
Premature ovarian failure, <i>n</i>	9	-	
Pernicious anaemia, <i>n</i>	4	1	
Hyperthyroidism, <i>n</i>	4	1	
Hypoparathyroidism, n	3	_	
Vitiligo, n	3	-	
Coeliac disease, <i>n</i>	2	1	
Alopecia, n	2	/	
Ulcerative colitis, <i>n</i>	1	1	
Multiple sclerosis, n	-	1	
Systemic lupus erythematosus, <i>n</i>	1	1	
Sarcoidosis, <i>n</i>	1	-	
Pyoderma gangrenosum, <i>n</i>	1	-	
Hypertension, n (%)	11 (13.4)	4 (5.0)	0.102
Asthma/COPD, n (%) <sup>a</sup>	11 (13.4)	3 (3.8)	0.047
Bone loss (osteopaenia/osteoporosis), n (%)	10 (12.2)	6 (7.5)	0.431
Reduced fertility, <sup>b</sup> <i>n</i> (%)	9 (11.0)	7 (8.8)	0.793
Cardiovascular events, <i>n</i> (%) <sup>a</sup>	8 (9.8)	1 (1.3)	0.034
Type-2 diabetes mellitus, $n$ (%)	5 (6.1)	4 (5.0)	>0.999
Sleep apnoea, <i>n</i> (%)	3 (3.7)	-	0.246
Depression, n (%)	3 (3.7)	6 (7.5)	0.325
History of cancer, <i>n</i> (%)	2 (2.4)	1 (1.3)	>0.999
Candidiasis, n (%)	1 (1.2)	-	>0.999
Testicular adrenal rest tumours, <i>n</i> (% of men)	_	7 (23.3)	
Recurrent lower respiratory tract infections, <i>n</i> (%)	-	2 (2.5)	0.242
Ehlers-Danlos syndrome, $n$ (%)	-	2 (2.5)	0.242
Charlson Comorbidity Index, n (%)			
0 <sup>a</sup>	28 (34.1)	57 (71.3)	<0.001
1	15 (18.3)	15 (18.8)	>0.999
≥2 <sup>a</sup>	39 (47.6)	8 (10.0)	<0.001
Glucocorticoid replacement therapy			
Immediate-release hydrocortisone, $n$ (%) <sup>a</sup>	80 (97.6)	36 (45.0)	<0.001
Prednisolone, <i>n</i> (%)ª	1 (1.2)	23 (28.8)	<0.001
Chronocort, <i>n</i> (%)	-	11 (13.8)	<0.001
Dexamethasone, <i>n</i> (%) <sup>a</sup>	-	6 (7.5)	0.013
Plenadren, <i>n</i> (%)	1 (1.2)	-	>0.999
Hydrocortisone + prednisolone, <i>n</i> (%)	-	1 (1.3)	0.494
Prednisolone + dexamethasone, <i>n</i> (%)	-	1 (1.3)	0.494
None, <i>n</i> (%)	-	2 (2.5)	0.242
Hydrocortisone-equivalent dose, median (IQR) <sup>c</sup>	20 (20–25)	24 (20–30)	0.328
Mineralocorticoid replacement, <i>n</i> (%) <sup>a</sup>	79 (96.3)	50 (68.5)	<0.001
Fludrocortisone dose, median (IQR) <sup>a</sup>	100 (100–188)	200 (119–200)	<0.001
DHEA replacement, <i>n</i> (% of women)	9 (17.6)	-	

<sup>a</sup>Significant figures; <sup>b</sup>includes patients with premature ovarian failure; <sup>c</sup>conversion factors: prednisolone 0.25; dexamethasone 0.025. BMI, body mass index; COPD, chronic obstructive pulmonary disease; DHEA, dehydroepiandrosterone; IQR, interquartile range.





Compared to patients with CAH, patients with AD were more likely to suffer from comorbid autoimmune diseases, cardiovascular disease and chronic respiratory disease (Table 1).

About 97.6% of patients with AD were treated with immediate-release hydrocortisone, compared to

45.0% of patients with CAH. Other steroid replacement therapies in patients with CAH included prednisolone (28.8%), modified-release hydrocortisone (13.8%) and dexamethasone (7.5%). The hydrocortisone-equivalent dosages did not differ significantly between the two groups. About 96.3% of patients with AD and 68.5% of

**Table 2** COVID-19 infection status and health during the preceding 12 months.

	Addison's disease	Congenital adrenal hyperplasia	
	( <i>n</i> = 82)	( <i>n</i> = 80)	P-value
COVID-19 diagnosed or suspected, n (%) <sup>a</sup>	15 (18.3) <sup>b</sup>	32 (40.0)	0.003
COVID-19 diagnosed	5	9	
Severe disease	2	2	
COVID-19 suspected but subject not tested	10	23	
Had sick days in the past 12 months, n (%)	51 (62.2) <sup>c</sup>	50 (62,5) <sup>c</sup>	>0.999
Malaise/fatigue	17	15	01000
COVID-19 (confirmed or suspected)	10	15	
Gastroenteritis	8	8	
Urinary tract infection	4	3	
Upper respiratory tract infection	-	10	
Lower respiratory tract infection	2	1	
Skin infaction	2	I	
	2	-	
Flu like symptoms	2	-	
Flu-like symptoms	3	2	
Psychological stress/pereavement	3	2	
Physical trauma	2	1	
COVID-19 vaccine/flu jab	2	2	
Ear infection	2	1	
Irritable bowel syndrome	-	3	
Physical activity	-	2	
Headache/migraine	-	2	
Cough	-	1	
Non-specific symptoms	-	1	
Had an adrenal crisis in the past 12 months, <i>n</i> (%)	10 (12.2)	8 (10.0) <sup>d</sup>	0.804
COVID-19 (confirmed or suspected)	2	2	
Gastroenteritis	2	1	
Urinary tract infection	2	-	
Dental abscess	1	-	
Flu-like symptoms	1	-	
Zoledronate infusion	1	-	
Unknown trigger	1	5	
Physical activity	/	1	
Had hospital admissions in the past 12 months, n (%) <sup>e</sup>	8 (9.8)	14 (17.5) <sup>d</sup>	0.174
Physical trauma/traumatic fracture	2	2	
Lower respiratory tract infection	_	3	
Gastroenteritis	1	1	
Surgery (planned or emergency)	1	2	
Transient ischaemic attack	-	2	
Urinary tract infection	_	2	
Chest pain	1	-	
Hypertension tachycardia	1	_	
Miscarriage	1	_	
Delivery	-	- 1	
Diabatic katoacidosis	-	ı 1	
	-	I	
Cellulius	I	-	
wigranie	-	I	

<sup>a</sup>Significant figure; <sup>b</sup>of the three patients with autoimmune polyendocrine syndrome type 1, one developed mild symptoms of COVID-19 and did not require hospital admission; <sup>c</sup>some patients had multiple sick days; <sup>d</sup>one patient had two adrenal crises and three hospital admissions; <sup>e</sup>excluding hospital admissions for adrenal crisis and/or severe COVID-19.

© 2023 the author(s) Published by Bioscientifica Ltd



This work is licensed under a Creative Commons Attribution 4.0 International License. aded from Bioscientifica.com at 08/09/2023 09:49:47AM via free access





#### Figure 2

Prevalence of COVID-19 in the study participants. Forty-seven patients with primary adrenal insufficiency (PAI) had suspected or confirmed COVID-19 based on symptoms or a positive polymerase chain reaction (PCR) test. Patients with congenital adrenal hyperplasia (CAH) had significantly higher odds of COVID-19 compared to those with Addison's disease (AD).

patients with CAH were treated with mineralocorticoid replacement therapy, with the latter group receiving a higher daily dose of fludrocortisone (Table 1).

#### **Illness during the COVID-19 pandemic**

Forty-seven patients (29.0%) were suspected to have had COVID-19 by the time the questionnaire was completed (Table 2 and Fig. 2). Only a minority were tested, and the infection was confirmed by PCR test in five patients with AD and nine with CAH. Severe COVID-19 affected two patients with AD and two with CAH (Appendix Table 1). No deaths were recorded.

Patients with CAH were more likely to have had a diagnosis of confirmed or clinically suspected COVID-19 (40.0% vs 18.3% of patients with AD; unadjusted odds ratio 2.98 (95% CI 1.48-6.23), P < 0.001). This remained significant after adjustment for age, sex, BMI, type of glucocorticoid used, daily hydrocortisone-equivalent dose, fludrocortisone treatment, Charlson Comorbidity Index score, smoking status, employment, living arrangements and ethnicity (adjusted odds ratio 2.53 (95% CI 1.07-6.16), P = 0.036). In line with the higher odds observed in patients with CAH, the 47 patients with PAI and suspected or confirmed COVID-19 were less commonly treated with immediate-release hydrocortisone or fludrocortisone and had a lower burden of comorbidity (Appendix Table 2). Subgroup analyses in patients with AD showed that those with previously suspected or confirmed COVID-19 were younger, had a lower cardiometabolic burden and were more likely to work at the time of the interview (Appendix Table 3). Patients with CAH and previously suspected or confirmed COVID-19 were older and less likely to be

treated with mineralocorticoid-replacement therapy (Appendix Table 4).

Within the preceding 12 months of interview, 62.3% of patients with PAI had to follow sick day rule advice for intercurrent illness, with suspected or confirmed COVID-19 being the second most common reason after general malaise/fatigue (18.5% and 23.7% of all sick day events, respectively). During the year preceding the interview, 18 patients with PAI (11.1%) suffered at least one adrenal crisis and suspected or confirmed COVID-19 was the leading trigger where a trigger could be identified, with four cases reported (Table 2).

# COVID-19 risk factors and health promotion attitudes

With regards to social risk factors for developing COVID-19 enquired about in the interviews (smoking status, employment and living arrangements), there was no significant difference between patients with AD or CAH (Table 3).

More patients with AD had received, or planned to receive, the COVID-19 vaccine compared to patients with CAH (96.3% and 80.0%, respectively; P < 0.001). More patients with AD had also received, or planned to receive, the flu vaccine during the 2020–21 period (75.6% vs 51.3%; P < 0.002) and reported receiving it every year (73.2% vs 52.5%; P < 0.009). All patients with AD were aware of sick day rules, compared to 93.8% of patients with CAH. Most patients with PAI had a steroid emergency card (89.5%) and a hydrocortisone self-injection kit available (90.1%) at the time of the interview, but patients with AD were more likely to have received self-injection training (91.5%)





	Addison's disease (n = 82)	Congenital adrenal hyperplasia $(n = 80)$	P-value
Smoking habit, n (%)			
Non-smoker	65 (79.3)	70 (87.5)	0.206
Former smoker	8 (9.8)	6 (7.5)	0.781
Current smoker	9 (11.0)	4 (5.0)	0.247
Currently working, <i>n</i> (% of currently working)	34 (41.5)	35 (43.8)	0.874
Key workers <sup>a</sup>	10 (29.4)	18 (51.4)	0.087
Able to work from home	17 (50.0)	16 (45.7)	0.811
Household arrangement, <i>n</i> (%)			
Lives alone	14 (17.1)	9 (11.3)	0.369
Lives with another person	35 (42.7)	31 (38.8)	0.635
Lives with more than one person	33 (40.2)	33 (41.3)	>0.999
Has the NHS COVID-19 app, <i>n</i> (%)	33 (40.2)	41 (51.3)	0.207
Had/planning to have the COVID-19 vaccine, <i>n</i> (%) <sup>b</sup>	79 (96.3)	64 (80.0)	0.001
Had/planning to have the flu vaccine this year, n (%) <sup>b</sup>	62 (75.6)	41 (51.3)	0.002
Has the flu vaccine every year, $n$ (%) <sup>b</sup>	60 (73.2)	42 (52.5)	0.009
Is aware of the sick day rules, $n$ (%) <sup>b</sup>	82 (100)	75 (93.8)	0.028
Has a steroid emergency card, n (%)	75 (91.5)	70 (87.5)	0.452
Always carries the card	63 (84.0)	62 (88.6)	0.477
Carries the card just occasionally	10 (13.3)	5 (7.1)	0.280
Never carries the card	2 (2.7)	5 (7.1)	0.263
Has a steroid injection kit, <i>n</i> (%)	76 (92.7)	70 (87.5)	0.303
It is up to date	54 (71.1)	43 (61.4)	0.226
It is out of date	19 (25.0)	22 (31.3)	0.462
Does not know whether it is still in date	3 (3.9)	5 (7.1)	0.481
Has received training for self-injection, <i>n</i> (%) <sup>b</sup>	75 (91.5)	64 (80.0)	0.044
Family members have received injection training, <i>n</i> (%)	54 (65.9)	54 (67.5)	0.869
Is confident in self-injecting, n (%)	54 (65.9)	43 (53.8)	0.149
Has ever self-injected, n (%)	17 (20.7)	17 (21.3)	>0.999
Wears medical alert jewellery, n (%) <sup>b</sup>	53 (64.6)	29 (36.3)	0.001

<sup>a</sup>Occupation groups: health and social care, education and childcare, food and other necessary goods; key public services; local and national government; utilities and communication; public safety and national security; transport; <sup>b</sup>significant figures. NHS, National Health Service.

vs 80.0% of patients with CAH; P < 0.044). Medical alert jewellery was worn more frequently by patients with AD (64.6% vs 36.3%; 0.001). Less than half of the patients included in the study reported having the National Health Service COVID-19 application on their phones (AD, 40.2%; CAH, 51.3%) (Table 3).

# Discussion

In this cross-sectional study, we demonstrated that COVID-19 has been a major precipitant for ill health in a large cohort of patients with PAI, commonly leading to sick day events and even culminating in hospital admission for adrenal crisis in four patients. We found higher odds of suspected or confirmed COVID-19 among patients with CAH compared to those with AD.

Infections, and most importantly those of the respiratory system, are an established cause of poor health



outcomes in patients with PAI, resulting in increased

morbidity and mortality (5, 7, 8, 9). Studies have found



**12**:8

e230122



to mount a sufficient acute response to infection, thus increasing the likelihood of complications (22, 23). Another mechanism that has been proposed could be dehydroepiandrosterone (DHEA) deficiency in patients with AD, since DHEA may play a role in regulating immune function (9, 24). Only 9 women with AD in our study were treated with DHEA replacement; hence, we could not carry out a subgroup analysis of the impact of this treatment on COVID-19 risk.

We found that 4 out of 47 patients with suspected or confirmed COVID-19 (8.5%) were admitted to the hospital for severe disease. Our results are in line with a recently published study by Öster S *et al.* (25); in a large Swedish cohort of patients with AD, the authors found that 17% of patients reported a diagnosis of COVID-19, of which 8.5% required hospital admission (25). An analysis of the European Registries for Rare Endocrine Conditions found a higher risk of hospital admission (14%) in a group of 57 patients with adrenal insufficiency (26).

Several agencies have released guidelines to advise both clinicians and patients as to the risks associated with COVID-19 in PAI (3, 23, 27, 28, 29). Advice has primarily relied on expert opinion, owing to the rarity of original clinical data. Due to concerns over a higher risk of complications and mortality, the European Society of Endocrinology advises that sick day rules, where patients at least double steroid replacement therapy during periods of physiological stress, should be implemented immediately in suspected or confirmed COVID-19, even if symptoms are only mild. Patients are also recommended to ensure they have adequate supplies of replacement therapy at home, in case of further national lockdowns or the requirement for self-isolation (29). Other guidelines emphasise the importance of empowering patients to take a proactive stance towards their own health promotion during the pandemic, through means such as education around social distancing and discussion of current COVID-19 guidelines (3).

Our data highlight differences in COVID-19 rates, and attitudes towards illness prevention, between patients with AD and CAH. The latter were more likely to develop suspected or confirmed COVID-19 during the study period, which is in line with a recent study by Yedinak C and Ross IL (30). The authors conducted a multinational survey on patients with a self-reported diagnosis of adrenal insufficiency and found that 8 of 72 patients with CAH (11.1%) tested positive for COVID-19 as compared to 32 of 1219 patients with other adrenal insufficiency aetiologies (2.6%) (30). The authors also found that COVID-19-positive patients were treated with higher daily hydrocortisone replacement doses, whilst we did not observe this.

While patients with CAH experienced higher rates of suspected or confirmed COVID-19 in our study, they also demonstrated reduced self-protective attitudes and engagement with healthcare services. For example, 80% of patients with CAH had received, or planned on receiving, the COVID-19 vaccine at the time of the interview, compared to almost all patients with AD. With regards to the flu vaccine, only about half of patients with CAH reported receiving it every year, compared to around three-quarters of those with AD. Additionally, patients with CAH were significantly less likely to have received training in self-injection or to wear medical alert jewellery. Prior work has identified problems in transitioning patients with CAH from paediatric to adult endocrine services, with patients often being lost to follow-up (31, 32, 33). In 2010, it was found that less than 10% of the expected UK adult population of patients with CAH were under specialist endocrine care (32). Considering this, the reduced self-protective attitudes observed in this study are perhaps unsurprising. It is imperative to ensure that patients with PAI, especially those with CAH, are having regular follow-up appointments with some focus on health promotion education and reinforcement of selfmanagement skills.

Besides reduced engagement with healthcare services, our results highlight other areas that may contribute to increased rates of COVID-19 among patients with CAH. Patients with AD were older and had a greater burden of comorbidity than those with CAH. It may be that patients with AD perceived themselves to be at a greater risk of falling unwell with COVID-19 for these reasons and therefore felt more inclined to follow our shielding advice stringently. It is also worth appreciating our data on employment. Although a similar number of patients with CAH and AD were working at the time of the interview, 51.4% of employed patients with CAH were key workers who continued to work as normal during the pandemic, compared to only 29.4% of those with AD. These patients likely faced higher levels of exposure to COVID-19.

This study benefits from its large cohort of wellcharacterised patients with PAI, with a sample size large enough to enable comparisons between CAH and AD. Additionally, the use of a semi-structured questionnaire allowed the collection of comprehensive qualitative data and a thorough exploration of patients' health attitudes. However, the present study also has some limitations. First, we relied on patient-reported suspicion of COVID-19 for many cases, as PCR testing in the UK was not





available out-of-hospital to the wider public until mid-2020 (34). A patient's interpretation of symptoms could be inaccurate, resulting in discrepancies between suspected and confirmed cases of COVID-19. Second, we did not perform a case-control or population-based analysis with matched subjects without PAI. This would have helped in determining whether patients with PAI truly carried a higher risk of COVID-19 compared to people with normal adrenal function. Finally, the single-centre nature of this study limits the generalisability of our findings.

In conclusion, we found that patients with CAH had greater odds of contracting COVID-19 than patients with AD and reduced self-protective healthcare attitudes have likely contributed to this. Patients with PAI must understand their heightened susceptibility to infection and the importance of self-protective attitudes to prevent health deterioration.

#### Supplementary materials

This is linked to the online version of the paper at https://doi.org/10.1530/ EC-23-0122.

#### **Conflict of Interest**

The authors do not declare a conflict of interest in relation to this work.

#### Funding

L.M.S. is the recipient of a Health Education England/National Institute for Health and Care Research (HEE/NIHR) doctoral fellowship (grant reference number ICA CDRF-2018-04-ST2-050). A.P. is the recipient of a Diabetes UK Sir George Alberti Research Training (grant reference number 18/0005782). A.P. receives support from the NIHR Birmingham Biomedical Research Centre at the University Hospitals Birmingham NHS Foundation Trust and the University of Birmingham (grant reference number NIHR203326). The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care UK.

#### Data availability

De-identified, individual participant-level data are available upon reasonable request. All requests should be sent to the corresponding author.

### References

- 1 Bornstein SR, Allolio B, Arlt W, Barthel A, Don-Wauchope A, Hammer GD, Husebye ES, Merke DP, Murad MH, Stratakis CA *et al.* Diagnosis and treatment of primary adrenal insufficiency: an Endocrine Society clinical practice guideline. *Journal of Clinical Endocrinology and Metabolism* 2016 **101** 364–389. (https://doi. org/10.1210/jc.2015-1710)
- 2 Neary N & Nieman L. Adrenal insufficiency: etiology, diagnosis and treatment. *Current Opinion in Endocrinology, Diabetes, and Obesity* 2010
   17 217–223. (https://doi.org/10.1097/MED.0b013e328338f608)

- 3 Arlt W, Baldeweg SE, Pearce SHS & Simpson HL. Endocrinology in the time of COVID-19: management of adrenal insufficiency. *European Journal of Endocrinology* 2020 **183** G25–G32. (https://doi.org/10.1530/EJE-20-0361)
- 4 Ngaosuwan K, Johnston DG, Godsland IF, Cox J, Majeed A, Quint JK, Oliver N & Robinson S. Increased mortality risk in patients with primary and secondary adrenal insufficiency. *Journal of Clinical Endocrinology and Metabolism* 2021 **106** e2759–e2768. (https://doi. org/10.1210/clinem/dgab096)
- 5 Bergthorsdottir R, Leonsson-Zachrisson M, Oden A & Johannsson G. Premature mortality in patients with Addison's disease: a populationbased study. *Journal of Clinical Endocrinology and Metabolism* 2006 **91** 4849–4853. (https://doi.org/10.1210/jc.2006-0076)
- 6 Puar TH, Stikkelbroeck NM, Smans LC, Zelissen PM & Hermus AR. Adrenal Crisis: still a deadly event in the 21st century. *American Journal of Medicine* 2016 **129** 339.e1–339.e9. (https://doi.org/10.1016/j. amjmed.2015.08.021)
- 7 Falhammar H, Frisen L, Norrby C, Hirschberg AL, Almqvist C, Nordenskjold A & Nordenstrom A. Increased mortality in patients with congenital adrenal hyperplasia due to 21-hydroxylase deficiency. *Journal of Clinical Endocrinology and Metabolism* 2014 **99** E2715–E2721. (https://doi.org/10.1210/jc.2014-2957)
- 8 Erichsen MM, Lovas K, Fougner KJ, Svartberg J, Hauge ER, Bollerslev J, Berg JP, Mella B & Husebye ES. Normal overall mortality rate in Addison's disease, but young patients are at risk of premature death. *European Journal of Endocrinology* 2009 **160** 233–237. (https://doi. org/10.1530/EJE-08-0550)
- 9 Tresoldi AS, Sumilo D, Perrins M, Toulis KA, Prete A, Reddy N, Wass JAH, Arlt W & Nirantharakumar K. Increased infection risk in Addison's disease and congenital adrenal hyperplasia. *Journal of Clinical Endocrinology and Metabolism* 2020 **105** 418–429. (https://doi. org/10.1210/clinem/dgz006)
- 10 Hahner S, Spinnler C, Fassnacht M, Burger-Stritt S, Lang K, Milovanovic D, Beuschlein F, Willenberg HS, Quinkler M & Allolio B. High incidence of adrenal crisis in educated patients with chronic adrenal insufficiency: a prospective study. *Journal of Clinical Endocrinology and Metabolism* 2015 **100** 407–416. (https://doi. org/10.1210/jc.2014-3191)
- 11 Hahner S, Loeffler M, Bleicken B, Drechsler C, Milovanovic D, Fassnacht M, Ventz M, Quinkler M & Allolio B. Epidemiology of adrenal crisis in chronic adrenal insufficiency: the need for new prevention strategies. *European Journal of Endocrinology* 2010 162 597–602. (https://doi.org/10.1530/EJE-09-0884)
- 12 Rushworth RL & Torpy DJ. A descriptive study of adrenal crises in adults with adrenal insufficiency: increased risk with age and in those with bacterial infections. *BMC Endocrine Disorders* 2014 **14** 79. (https://doi.org/10.1186/1472-6823-14-79)
- 13 Goldman JD, Robinson PC, Uldrick TS & Ljungman P. COVID-19 in immunocompromised populations: implications for prognosis and repurposing of immunotherapies. *Journal for ImmunoTherapy of Cancer* 2021 **9**. (https://doi.org/10.1136/jitc-2021-002630)
- 14 Bancos I, Hazeldine J, Chortis V, Hampson P, Taylor AE, Lord JM & Arlt W. Primary adrenal insufficiency is associated with impaired natural killer cell function: a potential link to increased mortality. *European Journal of Endocrinology* 2017 **176** 471–480. (https://doi.org/10.1530/EJE-16-0969)
- 15 Carosi G, Morelli V, Del Sindaco G, Serban AL, Cremaschi A, Frigerio S, Rodari G, Profka E, Indirli R, Mungari R *et al*. Adrenal Insufficiency at the Time of COVID-19: A Retrospective Study in Patients Referring to a Tertiary Center. *Journal of Clinical Endocrinology and Metabolism* 2021 **106** e1354–e1361. (https://doi.org/10.1210/clinem/dgaa793)
- 16 Martino M, Aboud N, Cola MF, Giancola G, Ciarloni A, Salvio G & Arnaldi G. Impact of COVID-19 pandemic on psychophysical stress in patients with adrenal insufficiency: the CORTI-COVID study. *Journal* of Endocrinological Investigation 2021 44 1075–1084. (https://doi. org/10.1007/s40618-020-01422-2)





- 17 Li D, Suresh M, Abbondanza T, Vaidya A & Bancos I. The impact of the COVID-19 pandemic on self-reported outcomes in patients with adrenal insufficiency. *Journal of Clinical Endocrinology and Metabolism* 2021 **106** e2469–e2479. (https://doi.org/10.1210/clinem/dgab334)
- 18 Charlson ME, Pompei P, Ales KL & MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. *Journal of Chronic Diseases* 1987 **40** 373–383. (https://doi.org/10.1016/0021-9681(87)90171-8)
- 19 Allolio B. Extensive expertise in endocrinology. Adrenal crisis. *European Journal of Endocrinology* 2015 **172** R115–R124. (https://doi. org/10.1530/EJE-14-0824)
- 20 Smans LC, Souverein PC, Leufkens HG, Hoepelman AI & Zelissen PM. Increased use of antimicrobial agents and hospital admission for infections in patients with primary adrenal insufficiency: a cohort study. *European Journal of Endocrinology* 2013 **168** 609–614. (https:// doi.org/10.1530/EJE-12-0879)
- 21 Bjornsdottir S, Sundstrom A, Ludvigsson JF, Blomqvist P, Kampe O & Bensing S. Drug prescription patterns in patients with Addison's disease: a Swedish population-based cohort study. *Journal of Clinical Endocrinology and Metabolism* 2013 **98** 2009–2018. (https://doi. org/10.1210/jc.2012-3561)
- 22 Teblick A, Peeters B, Langouche L & Van den Berghe G. Adrenal function and dysfunction in critically ill patients. *Nature Reviews*. *Endocrinology* 2019 **15** 417–427. (https://doi.org/10.1038/s41574-019-0185-7)
- 23 Isidori AM, Arnaldi G, Boscaro M, Falorni A, Giordano C, Giordano R, Pivonello R, Pofi R, Hasenmajer V, Venneri MA *et al.* COVID-19 infection and glucocorticoids: update from the Italian Society of Endocrinology Expert Opinion on steroid replacement in adrenal insufficiency. *Journal of Endocrinological Investigation* 2020 **43** 1141–1147. (https://doi.org/10.1007/s40618-020-01266-w)
- 24 Svec F & Porter JR. The actions of exogenous dehydroepiandrosterone in experimental animals and humans. *Proceedings of the Society for Experimental Biology and Medicine* 1998 **218** 174–191. (https://doi. org/10.3181/00379727-218-44285)
- 25 Oster S, Esposito D, Aranda-Guillen M, Akerman AK, Wahlberg J, Husebye ES, Kampe O, Botusan IR, Dahlqvist P, Bergthorsdottir R *et al.* Self-management and hospitalization in 615 Swedish patients with Addison's disease during the coronavirus disease 2019 pandemic:

a retrospective study. *European Journal of Endocrinology* 2023 **188**. (https://doi.org/10.1093/ejendo/lvad010)

- 26 Nowotny HF, Bryce J, Ali SR, Giordano R, Baronio F, Chifu I, Tschaidse L, Cools M, van den Akker EL, Falhammar H *et al.* Outcome of COVID-19 infections in patients with adrenal insufficiency and excess. *Endocrine Connections* 2023 **12** e220416. (https://doi. org/10.1530/EC-22-0416)
- 27 Pal R & Bhadada SK. Managing common endocrine disorders amid COVID-19 pandemic. *Diabetes and Metabolic Syndrome* 2020 14 767–771. (https://doi.org/10.1016/j.dsx.2020.05.050)
- 28 Chifu I, Detomas M, Dischinger U, Kimpel O, Megerle F, Hahner S, Fassnacht M & Altieri B. Management of patients with glucocorticoidrelated diseases and COVID-19. *Frontiers in Endocrinology (Lausanne)* 2021 **12** 705214. (https://doi.org/10.3389/fendo.2021.705214)
- 29 Puig-Domingo M, Marazuela M & Giustina A. COVID-19 and endocrine diseases. A statement from the European Society of Endocrinology. *Endocrine* 2020 68 2–5. (https://doi.org/10.1007/s12020-022-02294-5)
- 30 Yedinak C & Ross IL. Significant risk of COVID-19 and relatedhospitalization among patients with adrenal insufficiency: A large multinational survey. *Frontiers in Endocrinology (Lausanne)* 2022 **13** 1042119. (https://doi.org/10.3389/fendo.2022.1042119)
- 31 Gleeson H, Davis J, Jones J, O'Shea E & Clayton PE. The challenge of delivering endocrine care and successful transition to adult services in adolescents with congenital adrenal hyperplasia: experience in a single centre over 18 years. *Clinical Endocrinology* 2013 **78** 23–28. (https://doi.org/10.1111/cen.12053)
- 32 Arlt W, Willis DS, Wild SH, Krone N, Doherty EJ, Hahner S, Han TS, Carroll PV, Conway GS, Rees DA *et al.* Health status of adults with congenital adrenal hyperplasia: a cohort study of 203 patients. *Journal of Clinical Endocrinology and Metabolism* 2010 **95** 5110–5121. (https:// doi.org/10.1210/jc.2010-0917)
- 33 Jenkins-Jones S, Parviainen L, Porter J, Withe M, Whitaker MJ, Holden SE, Morgan CL, Currie CJ & Ross RJM. Poor compliance and increased mortality, depression and healthcare costs in patients with congenital adrenal hyperplasia. *European Journal of Endocrinology* 2018 **178** 309–320. (https://doi.org/10.1530/EJE-17-0895)
- 34 Briggs A, Jenkins D & Frasrer C. *NHS Test and Trace: the journey so far.* 2020. (available at: https://www.health.org.uk/publications/long-reads/nhs-test-and-trace-the-journey-so-far)

Received 2 April 2023 Accepted 26 May 2023 Available online 30 May 2023 Version of Record published 12 July 2023

